Author's response to reviews

Title: An epidemiological study assessing the prevalence of human papillomavirus types in women in the Kingdom of Bahrain

Authors:

Khairya Moosa (drkhairyamoosa@gmail.com)
Adel Alsayyad (asayyad@health.gov.bh)
Wim Quint (w.g.v.quint@ddl.nl)
Kusuma Gopola (kusuma.b.gopola@gsk.com)
Rodrigo DeAntonio (rodrigo.d.deantonio@gskbio.com)

Version: 8
Date: 17 November 2014

Author's response to reviews:

November 12, 2014

Dear Dr. Prof Sophie F Derchain,

Journal Editorial Office
BioMed Central

Please find enclosed a manuscript entitled “An epidemiological study assessing the prevalence of human papillomavirus types in women in the Kingdom of Bahrain” (Manuscript ID: 6792171812874568) which we are resubmitting for consideration by BMC Cancer.

Please also find attached itemized responses to the editorial comments. The authors have attempted to modify the manuscript as requested within the limits of the study objectives. Changes from the original submitted manuscript are highlighted in track changes.

We hope the revised manuscript is now acceptable for publication in BMC Cancer.

We look forward to hear the outcome of the review process.

On behalf of all authors,
Yours sincerely,

Dr. Khairya Moosa

Consultant Family Physician/ Assistant Professor

Address: Arabian Gulf University/Medical College, Bahrain

Email: drkhairyamoosa@gmail.com

Mobile No: +973-39628028